Author’s response to reviews

Title: Variations and inter-relationship in outcome from emergency admissions in England. A retrospective analysis of Hospital Episode Statistics from 2005-2010

Authors:

Peter J.E. Holt (peteholt@btinternet.com)
Sidartha Sinha (peteholt@btinternet.com)
Baris A Ozdemir (peteholt@btinternet.com)
Alan KArthikesalingam (peteholt@btinternet.com)
Jan D Poloniecki (peteholt@btinternet.com)
Matt M Thompson (peteholt@btinternet.com)

Version: 3 Date: 31 January 2014

Author’s response to reviews: see over
Author’s response to reviews

Title: Variations and inter-relationship in outcome from emergency admissions in England. A retrospective analysis of Hospital Episode Statistics from 2005-2010

Authors:

Peter J.E. Holt pholt@sgul.ac.uk
Sidartha Sinha sid261@hotmail.com
Baris A Ozdemir drbaozdemir@gmail.com
Alan Karthikesalingam alankarthi@gmail.com
Jan D Poloniecki sgp300@sgul.ac.uk
Matt M Thompson Matt.Thompson@stgeorges.nhs.uk

Version: 2 Date: 23rd January 2014

Author’s response to reviews: see over
Dear Biomed Central Editorial Team,

Please find attached the responses to your editorial office comments.

MS: 1882070134119445

1. Title page

Please include a title page at the front of your manuscript file. It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

This has been updated to include the information requested.

2. Name of ethics committee

Please update your ethics statement to include the name of the ethics committee that state that this study does not require ethical approval.

This has been done as requested. (WanREC)

3. Author's Contribution

For manuscripts with more than one author, all BMC Series journals require an Authors' Contributions section to be placed after the Competing Interests section.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. To qualify as an author one should 1) have made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not justify authorship.

We suggest the following format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies,
participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

Contributors who do not meet the criteria for authorship should be listed in an acknowledgements section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support.

An author’s contribution section has been added as requested.

Author’s Contribution:
PJEH: conception, design, analysis, drafting and critical revision
SS: conception, design, analysis, drafting and critical revision
BAO: design and drafting
AK: drafting and critical revision
JDP: Design, Analysis, Drafting and critical revision
MMT: Drafting, critical revision and guarantor

4. Please include your funding into the Acknowledgment Section. Place the Acknowledgement section after the Author’s Contribution.

The funding statement has been moved into an Acknowledgement section.

5. Please move the Competing Interest section after "Conclusion Section".

The competing interest statement has been moved to after the conclusion section as requested.